Author’s response to reviews

Title: Ulcerated calcification of the interventricular septum causing Transient Ischemic Attacks: Case Report

Authors:

Matthew Panagiotou (mspanag@otenet.gr)
Kostas Markakis (ekoletsis@hotmail.com)
Nikolaos Mourtzis (ekoletsis@hotmail.com)
Stella Economidis (ekoletsis@hotmail.com)
James Crockett (fccp_perfusionist@yahoo.co.uk)
Efstratios N Koletsis (ekoletsis@hotmail.com)

Version: 2 Date: 15 April 2007
Editors-in-Chief:

Vipin Zamvar,
Royal Infirmary of Edinburgh;

David Taggart,
Oxford Heart Centre, John Radcliffe Hospital

Athens, April 15th, 2007

Dear Editors,

We would like to thank you and the reviewers for reviewing our manuscript entitled: “Ulcerated calcification of the interventricular septum causing Transient Ischemic Attacks: Case Report” which was submitted for consideration for publication in the Journal of Cardiothoracic Surgery.

We have followed the request formatting changes as you can see in the attached response.

We are looking forward to hearing from you.

Yours Sincerely,

Matthew Panagiotou, MD, PhD.
Author for correspondence
Cardiothoracic Surgeon
Cardiac-Surgery Department, Athens Medical Centre
According to comments
Please remove the word counts, running title, and key words
Please remove the text: title page, institution, and the author qualifications.
Please put the author emails in a list beneath the corresponding author details
Please indicate the corresponding author with a symbol e.g. *
Structured Abstract -> Abstract
Please remove 'text of paper'
List of abbreviations -> Abbreviations
REFERENCES: -> References
Legends to figures -> Figure legends

Done, as requested